Author's response to reviews

Title: Exome sequencing of a patient with suspected mitochondrial disease reveals a likely multigenic etiology.

Authors:

William J Craigen (wcraigen@bcm.edu)
Brett H Graham (bgraham@bcm.edu)
Lee-Jun Wong (liw@bcm.edu)
Fernando Scaglia (fscaglia@bcm.edu)
Richard Alan Lewis (rlewis@bcm.edu)
Penelope E Bonnen (pbonnen@bcm.edu)

Version: 4 Date: 5 August 2013

Author's response to reviews: see over
August 5, 2013

RE: MS: 8564411098401832

Please find a revised version of the manuscript, “Exome sequencing of a patient with suspected mitochondrial disease reveals a likely multigenic etiology” that we would like you to consider for publication in BMC Medical Genetics. We have revised the manuscript to make it better comply with journal format and we have addressed the reviewer’s concerns.

If you have any questions, please contact me at pbonnen@bcm.edu.

Thank you,

Penelope E. Bonnen

Response to Editor’s Comments

The subject and his parents were consented for research and publication. We have included the following statement in the Methods section: “Informed consent was obtained from the subject and his parents for research and publication of clinical details according to a protocol approved by the Institutional Review Board for Baylor College of Medicine and Affiliated Hospitals.”

We have reformatted the references to match journal style.